Author’s response to reviews

Title: Antibacterial, Antioxidant and Immuno-modulatory Properties in extracts of Barleria Lupulina Lindl.

Authors:
RESHMA KUMARI (reshmagupta25@gmail.com)
Sumit Kumar (sumitarya360@gmail.com)
Ashwani Kumar (ashwanipharma03@gmail.com)
Kapil Goel (kapilgoel9@gmail.com)
Ramesh Dubey (profrcdubey@gmail.com)

Version: 2 Date: 30 Sep 2017

Author’s response to reviews:

Reviewer 4: Why methanol extract was the best?

Authors' response: Methanol is used for extraction various polar compounds but certain group of non polar compounds are fairly soluble in methanol. Therefore methanol is commonly used for extraction of bioactive or phytochemical compounds.